Author’s response to reviews

Title: Molecular prevalence of intestinal parasites infections in children with diarrhea in Franceville, southeast of Gabon.

Authors:

Sandrine Lydie Oyegue-Liabagui (lyds_ass@yahoo.fr)
Nal Kennedy Ndjangangoye (nndjangangoye@gmail.com)
Lady Charlene Kouna (charleneklc@gmail.com)
Gwladys Mirlande Lekolo (edmirlande2000@yahoo.fr)
Franck Mounioko (fmounioko@yahoo.fr)
Sylvie Kwedi Nolna (bellasylk@yahoo.com)
Jean-Bernard Lekana-Douki (lekana_jb@yahoo.fr)

Version: 6 Date: 01 May 2020

Author’s response to reviews:

Franceville, May 01, 2020

BMC Infectious Diseases
To the Editor-in-Chief

Dear Editor,

We would like to submit the revised manuscript entitled: “Molecular prevalence of intestinal parasites infections in children with diarrhea in Franceville, southeast of Gabon", by OYEGUE-LIABAGUI SANDRINE LYDIE et al., which we would like to present for an original research. The manuscript number for the first revision was ID: INFD-D-19-02707R4.

Indeed, after its third revision on April 16th, 2020, we received the comments of Editor on April 17th, 2020.

We have corrected the manuscript according to these comments and improved its quality.

We acknowledged all the comments of the editor and incorporated his suggestions. See from page 2 all our responses to the editor comments. All authors have seen and approved the manuscript and have taken due care to ensure both the integrity of this work and conformity with national and institutional policies.
Thank you in advance for taking the time to review this revised manuscript.

Sincerely yours,

Sandrine Lydie OYEGUE-LIABAGUI, corresponding author
lyds_ass@yahoo.fr

RESPONSE TO EDITOR’S COMMENTS

Editor Comments:

1. Please modify the current statement to include Sandrine Lydie OYEGUE-LIABAGUI, specifically mentioning them as the corresponding author. Until this is done, we cannot proceed with the manuscript.

Correct. The corresponding author has been indicated on the title page:
“* Correspondence: Sandrine Lydie Oyegue-Liabagui
lyds_ass@yahoo.fr
1Ecole Doctorale Régionale d’Afrique Centrale en Infectiologie Tropicale (ECODRAC),
Université des Sciences et Techniques de Masuku, BP 876 Franceville, Gabon.
2Unité d’Evolution, d’Épidémiologie et de Résistances Parasitaires (UNEEREP), Centre
International de Recherches Médicales de Franceville (CIRMF), B.P. 769 Franceville, Gabon”.

Title page

Declaration of competing interests:
We declare that we have no competing interests.